– 208 –

6. PUBLICATIONS AND REPORTS

6.1 Publications

P. Ackerbauer, P. Baumann, E.D. Bovet, W.H. Breunlich, T. Case, D. Chatellard, K.M. Crowe, H. Daniel, J.-P. Egger, F.J. Hartmann, E. Jeannet, M. Jeitler, P. Kammel, B. Lauss, K. Lou, V. Markushin, J. Marton, C. Petitjean, W. Prymas, W. Schott, R.H. Sherman, T. von Egidy, P. Wojciechowski, J. Zmeskal Muon transfer between excited states of muonic hydrogen isotopes PSI Nucl. Part. Phys. Newsl. 1991 (1991) 53

W. Assmann Ionization chambers for materials analysis with heavy ion beams Nucl. Instr. and Meth. B64 (1992) 267

K. Balog, K.E.G. Löbner, Th. Winkelmann Transputer als Front-End-Datenerfassungs- und Echtzeit-Analyse-System für Kernmassen-Messungen Springer Reihe Informatik Aktuell: Parallele Datenverarbeitung mit dem Transputer (1992) p. 304

J. Bartel, M.B. Johnson, M.K. Singham, W. Stocker Pion Inelastic Scattering and the Neutron Density in^{152}Sm Phys. Lett. B296 (1992) 5

G. Bartnitzky, H. Clement, M. Ermer, P. Grabmayr, G.J. Wagner, G. Graw, R. Hertenberger Model Unrestricted Analyses of Polarized Deuteron Scattering to Groundstates and Excited States Verhandl. DPG (VI) 27 (1992) 123 PD12

P. Baumann, H. Daniel, F.J. Hartmann, M. Mühlbauer, W. Schott, P. Wojciechowski, A. Fuchs, K. Lou, C. Petitjean, D. Taqqu, P. Hauser, F. Kottmann Measurement of the stopping power for μ^− and μ^+ at energies between 2 and 40 keV PSI Nucl. Part. Phys. Newsl. 1991 (1991) 59

T. Beha, W. Dünnweber, P. Klein, G. Lutz Performance of Silicon Trigger Counters in the Megarad Range IEEE Transactions Nucl. Sci. 39 (1992) 826

R. Behrisch, R. Grötzschel, E. Hentschel, W. Assmann HIERD analysis of the low-Z deposits on probes from the vessel walls of fusion experiments Nucl. Instr. and Meth. B68 (1992) 245

H. Berghammer, D. Vretenar, and P. Ring Relativistic Description of Coulomb Dissociation: a One-Dimensional Model Phys. Lett. 296B (1992) 290

– 209 –

D. Blamart, W. Rühm, W. Spiegel, K. Kato, G. Korschinek, H. Morinaga, G. Morteani, and E. Nolte Oxygen stable isotope measurements on a gravestone exposed to Hiroshima A-Bomb explosion and the Dosimetry System 1986 Chemical Geology (Isotope Geoscience Section) 101 (1992) 93

N. Blasi, S. Micheletti, M. Pignanelli, R. de Leo, R. Hertenberger, M. Bisenberger, D. Hofer, H. Kader, P. Schiemenz, G. Graw Study of^{110}Cd via the^{110}Cd(⃗d,t) Reaction Nucl. Phys. A536 (1992) 119

J. de Boer Do we Need more Physics? Do we Need more Physicists? Do we Need more Nuclear Physics? Status and Perspectives of Nuclear Energy: Fission and Fusion, 1992 CXVI Corso, Rendiconti della Società Italiana di Fisica, Bologna, Italy

J. de Boer, N.A. Gollwitzer, A. Lösch, H.J. Maier, H. Müller, M. Rohn, V. Grafen, N. Singh, B. Ackermann, T. Bihn, C. Günther, H. Hausmann, M. Marten-Tölle, R. Tölle In-beam Spectroscopy of^{231}Pa Z. Phys. A344 (1992) 41

B. Braun, H.G. Gaul, G. Graw, Minn-Tsong Lin, E. Steffens Measurement of Nuclear Spin Relaxation of Hydrogen on Coated Surfaces Verhandl. DPG (VI) 27 (1992) 37 E1.6

E.R. Crosson, R.K. Das, S.K. Lemieux, E.J. Ludwig, W.J. Thompson, M. Bisenberger, R. Hertenberger, D. Hofer, H. Kader, P. Schiemenz, G. Graw, A.M. Eiró, F.D. Santos Probing α-particle wave functions by (d,α) tensor analyzing powers Phys. Ref. C45(2) (1992)

J. Cub, U. Knopp, K.-H. Speidel, H. Busch, S. Kremeyer, H.-J. Wollersheim, N. Martin, X. Hong, K. Vetter, N.A. Gollwitzer, A.I. Levon, A. Booten Transient field measurements on^{40}Ar ions at the 1s electron Bohr velocity and the g-factor of the 2^+-state at 1.461 MeV Nucl. Phys. A549 (1992) 304

Crystal Barrel Collaboration The Crystal Barrel spectrometer at LEAR Nucl. Instr. and Meth. A321 (1992) 69

Crystal Barrel Collaboration Proton-antiproton annihilation into ηηπ-observation of a scalar resonance decaying into ηη Phys. Lett. B291 (1992) 347

Crystal Barrel Collaboration The pseudoscalar mixing angle θ_{ps} from η and η^′ production in p̄p annihilation at rest Phys. Lett. B294 (1992) 451

– 210 –

Crystal Barrel Collaboration P- versus S-wave annihilation annihilation at rest in LH_2 Phys. Lett. B297 (1992) 214

J. Czakanski, W. Zipper, M. Siemaszko, W. Dünnweber, W. Hering, D. Konnerth, W. Trombik, K.G. Bernhardt, H. Bohn, K.A. Eberhard, R. Vandenbosch Circular angular momentum of cluster evaporation:^{16}O(^{16}O,α)^{28}Si and^{16}O(^{16}O,^8Be)^{24}Mg Nucl. Phys. A542 (1992) 278

H. Daniel Muons as a probe in solid state and surface physics and in material research Nucl. Instr. and Meth. B68 (1992) 459

H. Daniel, F.J. Hartmann, M. Mühlbauer Monte-Carlo studies on beam-cooling by frictional forces PSI Nucl. Part. Phys. Newsl. (1991) 61

G. Dollinger, T. Faestermann, P. Maier-Komor High Resolution Depth Profiling of Light Elements Nucl. Instr. and Meth. B64 (1992) 422

K.-H. Ebeling, E. Hagn, E. Zech, A. Lerf Nuclear Orientation Study of PtTe_2 Phys. Rev. B45 (1992) 7474

T. von Egidy Statistical properties of complete level schemes in: ”Nuclear shapes and nuclear structure at low excitation energies” ed. M. Vergnes et al., Plenum Press, New York (1992) 373

D. Von-Eiff, S. Haddad and M.K. Weigel Approach to the relativistic Thomas-Fermi expansion for Green’s functions, phase-space densities, and densities Phys. Rev. C46 (1992) 230

D. Von-Eiff, M.K. Weigel Relativistic Thomas-Fermi calculations of hot nuclei Phys. Rev. C46 (1992) 1288 LBL-Report 32342

D. Von-Eiff and M.K. Weigel Relativistic Thomas-Fermi calculations of finit nuclei including quantum corrections Phys. Rev. C46 (1992) 1797

– 211 –

H. Elsenhans, L. Sehn, A. Faessler, H. Müther, N. Othsuka, H.H. Wolter Relativistic features of the effective interaction in heavy ion collisions Nucl. Phys. A536 (1992) 750

P. Fischer, G. Schütz, S. Scherle, M. Knülle, S. Stähler, G. Wiesinger Experimental Study of the Circular Magnetic X-Ray Dichroism in Ho-Metal, Ho_3Fe_5O_{12}, Ho_6Fe_{23} and Ho_2Co_{17} Solid State Comm. 82(11) (1992) 857

C. Fuchs, L. Sehn, H.H. Wolter Relativistic kinetic calculations of heavy ion collisions with phase space dependent fields Nucl. Phys. A545 (1992) 151c

N.K. Glendenning, F. Weber, and S.A. Moszkowski Neutron Stars in the Derivative Coupling Model Phys. Rev. C45 (1992) 844

R. Göbel, W. Arnold, Th. Frommhold, R. Stock, Th. Weber, U. Kneissl, F. Steiper, C. Kozhuharov, P. Kienle Bhabha to Møller Cross Section Ratios in the Energy Range 2.15 - 2.45 MeV GSI preprint 92-72 (1992)

R. Grötzschel, E. Hentschel, R. Klabes, U. Kreissig, C. Neelmeijer, W. Assmann, R. Behrisch Elemental analysis of thin layers by elastic heavy ion scattering Nucl. Instr. and Meth. B63 (1992) 77

E. Hagn New Developments and Trends in NMR-ON Hyp. Int. 75 (1992) 229

F.J. Hartmann Atomic capture of exotic particles in atoms, mixtures and compounds in: ”Exotic atoms in condensed matter”, ed. G. Benedek and H. Schneuwly Proceedings in Physics, Springer 59 (1992) 13

E. Hechtl An ion optical decel system for oblique angular impact of low-energy ions on targets Nucl. Instr. and Meth. in Phys. Res. B70 (1992) 441

E. Hechtl, W. Eckstein, J. Roth, and J. Lazlo Sputtering of tungsten by oxygen at temperatures up to 1900 K J. Nucl. Mater. 179 (1991) 290

– 212 –

T. Henkelmann and G. Korschinek Charge state distribution of tantalum ions produced simultaneously by CO_2 and Nd:YAG laser from a laser ion source Rev. Sci. Instrum. 63(4) (1992) 2828

W. Hering High Energy Gamma Rays Emitted by Compound Nuclei and Pre-Compound Nuclear Systems in: ”From Spectroscopy to Chaotic Features of Nuclear Systems”, D. Seeliger ed., World Scientific 1992

R. Hertenberger, D. Hofer, E. Müller-Zanotti, G. Graw, Gh. Cata-Danil, C. Hategan New Investigations on the Structure of the Nucleus^{110}Pd Verhandl. DPG (VI) 27 (1992) 99 F4.3

R.R. Hilton On the Coexistence of Collective and Single Particle Aspects of the Giant Angle Dipole Annals of Physics 214 (1992) 258-292

G. Hinderer, C. Rieß, U. Trinks Transverse Beam Dynamics and Coupling Effects in the TRITRON Nucl. Instr. and Meth. A317 (1992) 13

D. Hofer, P.F. Bortignon, Gh. Cata-Danil, C. Hategan, R. Hertenberger, E. Müller-Zanotti, G. Graw Zweistufige Anregungen in der unelastischen Streuung leichter Ionen Verhandl. DPG (VI) 27 (1992) 83 G3.9

H. Hofmann, D. Kiderlen and I. Tsekhmistrenko Statistical fluctuations for Landau’s kinetic equation Z. Phys. A341 (1992) 181

P. Hofmann, Y.S. Kim, H. Daniel, T. von Egidy, T. Haninger, F.J. Hartmann, P. David, H. Machner, G. Riepe, H.S. Plendl, K. Ziock, B. Wright, D. Bowman, W. Lynch, J. Lieb, J. Jastrzebski, W. Kurcewicz Antiproton induced fission and fragmentation Sov. J. Nucl. Phys. 55 (1992) 713 Original version in Yadernaya Fizika 55 (1992) 1281

H. Hofmann, S. Yamaji and A.S. Jensen Strength distribution of isoscalar vibrations around thermal equilibrium Phys. Lett. B286 (1992) 1

J. Homolka, W. Schott, W. Wagner, W. Wilhelm, M. Saber, R.E. Segel, R.D. Bent, M. Fatyga, R.E. Pollock, P. Kienle, K.E. Rehm Measurement of the^{12}C(p,π^o)^{13}N reaction by recoil detection Phys. Rev. C45 (1992) 1276

– 213 –

D.B. Ion, W. Stocker Nuclear Gamma Cherenkov Radiation from Charged Projectiles as a Coherent Effect in Nuclear Media Ann. Phys. (N.Y.) 213 (1992) 355

M. Jung, F. Bosch, K. Beckert, H. Eickhoff, H. Folger, B. Franzke, A. Gruber, P. Kienle, O. Klepper, W. Koenig, C. Kozhuharov, R. Mann, R. Moshammer, F. Nolden, U. Schaaf, G. Soff, P. Spädtke, M. Steck, Th. Stöhlker, K. Sümmerer First Observation of Bound-State β^− Decay Phys. Rev. Lett. 69 (1992) 2164

B. Kamys, Z. Rudy, J. Kisiel, E. Kwasniewicz, H.H. Wolter Simultaneous and sequential transfer of proton and alpha-particle in elastic^{11}B+^{16}O scattering Z. Phys. A342 (1992) 149

P. Kienle First Results from the Heavy Ion Synchrotron-Storage Cooler Ring Combination SIS/ESR at GSI GSI preprint 92-43 (1992)

D. Kiderlen, H. Hofmann and F.A. Ivanyuk Dynamical aspects of thermal nuclear properties Nucl. Phys. A550 (1992) 473

P. Kienle Latest Hot Topics in Atomic Physics Research at GSI GSI preprint 92-60 (1992)

G. Korschinek and T. Henkelmann Negative ions generated by laser-material interaction Rev. Sci. Instr. 63 (1992) 2672

H. Lenske Pair Correlations and the Structure of Exotic Nuclei Proc. Int. Symp. on Structure and Reactions of Unstable Nuclei, Niigata (Japan), June 1991 eds. K. Ikeda and Y. Suzuki, World Scientific, Singapore 1992

H. Lenske, H.H. Wolter Statistical direct reaction theory for dissipative heavy ion collisions in the Fermi energy domain Nucl. Phys. A538 (1992) 483c

A.I. Levon, J. de Boer, A.A. Pasternak, D.A. Volkov M1-transitions and the mutual orientation of the unpaired nucleons in odd-odd nuclei Z. Phys. A343 (1992) 131

– 214 –

A. Lösch, J. de Boer, N.A. Gollwitzer, A.I. Levon, G. Graw, R. Hertenberger, H. Kader, V. Grafen, B. Ackermann, T. Bihn, C. Günther, R.S. Simon Spectroscopy of^{229}Pa Verhandl. DPG (VI) 27 (1992) 211 F7.7

H. Machner, Sa Jun, G. Riepe, D. Protic, H. Daniel, T. von Egidy, F.J. Hartmann, W. Kanert, W. Markiel, H.S. Plendl, K. Ziock, R. Marshall, J.J. Reidy Fission fragment distribution following antiproton absorption at rest on^{238}U Z. Phys. A343 (1992) 73

J.P. Maharana, Y.K. Gambhir, J.A. Sheikh, and P. Ring Shape Coexistence and Extreme Deformations Near A=80 Phys. Rev. C46 (1992) R1163

S. Michaelsen, A. Harder, K.P. Lieb, G. Graw, D. Hofer, R. Hertenberger, P. Schiemenz, E. Müller- Zanotti, S. Robinson, A.P. Williams Spectroscopy of^{90}Y via the^{89}Y(n,γ ) and^{89}Y(d,p) reaction Verhandl. DPG (VI) 27 (1992) 43 G1.2

D. Müller, B. Bante, T. Faestermann, A. Gillitzer, H.-J. Körner, G. Korschinek, U. Bittner Accelerator Mass Spectrometry with a gas-filled Magnetic Spectrograph Nucl. Instr. and Meth. B68 (1992) 313

S. Neumaier, E. Nolte and A. Urban Neutrino oscillations in the sun and the^{205}Tl solar neutrino experiment Z. Phys. A341 (1992) 239

A.N. Ostrowski, H.G. Bohlen, A.S. Demyanova, B. Gebauer, R. Kalpakchieva, Ch. Langner, H. Lenske, H. von Lucke-Petsch, W. von Oertzen, A.A. Ogloblin, Y.E. Penionzhkevich, M. Wilpert, Th. Wilpert Mass Spectroscopy of^{13}Be Z. Phys. A343 (1992) 489

R. Rossignoli, P. Ring, and N. Dang Finite Temperature Projected Calculations in the Static Path Approximation Phys. Lett. 297B (1992) 9

W. Rühm, K. Kato, G. Korschinek, H. Morinaga and E. Nolte 36Cl and^{41}Ca depth profiles in a Hiroshima granite stone and the Dosimetry System 1986 Z. Phys. A341 (1992) 235

– 215 –

I. Romanski, I. Berkes, D.E. Brown, M. Dejesus, R. Eder, I.S. Grant, E. Hagn, P. Harding, P. Herzog, B. Hinfurtner, B. Kastelein, H. Postma, J. Prinz, P. Richards, K. Schlösser, N.J. Stone, L. Vanneste, E. Zech and the NICOLE and ISOLDE collaborations Time-Resolved and Time-Integral On-Line Nuclear Orientation Measurements of Neutron-Deficient Hg- Au-Pt-Ir Nuclei Hyp. Int. 75 (1992) 457

P. Schiemenz Polarized ion sources Rev. Sci. Instrum. 63(4) (1992) 2519-2524

P.Schiemenz, A. Ross, G. Graw Permanent sextupole magnets for polarized Stern Gerlach type atomic beam sources Rev. Sci. Instrum. 63(4) (1992) 2634-2635

G. Schütz Magnetic Photoabsorption with Circularly Polarized X-Rays Synchrotron Radiation: Selected Experiments in Condensed Matter Physics, Monte Veritá Birkhäuser Verlag Basel (1992) 25

G. Schütz, H. Ebert, P. Fischer, S. Rüegg, W.B. Zeper Spin Polarization of Pt in Pt/Co Multilayers studied by X-ray Absorption Mat. Res. Soc. Symp. Proc. 231 (1992) 77

G. Schütz, P. Fischer Circularly Polarized X-Rays Probing Nuclear Magnetic Moments and Magnetism of Solids Z. Phys. A341 (1992) 227

M.M. Sharma and P. Ring Neutron Skin of Spherical Nuclei in Relativistic and Non-Relativistic Mean-Field Approaches Phys. Rev. C45 (1992) 2514

M.M. Sharma and P. Ring Relativistic Mean-Field Description of Neutron-Deficient Platinum Isotopes Phys. Rev. C46 (1992) 1715

A. Shumshurov, G. Belajev, V. Dubenkov, A. Golubev, T. Henkelmann, G. Korschinek, B. Wolf, D.H.H. Hoffmann, B. Sharkov Laser plasma ion source for intense heavy ion beams Rev. Sci. Instrum. 63(4) (1992) 2834

K.-H. Speidel, J. Cub, W. Karle, U. Reuter, H. Busch, S. Kremeyer, J. Gerber Transient magnetic field under heavy ion bombardment of ferromagnetic solids Materials Sci. Forum 97-99 (1992) 615

– 216 –

K.-H. Speidel, J. Cub, U. Reuter, W. Karle, H. Busch, S. Kremeyer, J. Gerber, F. Hagelberg New determination of the magnetic moment of the^{54}Fe(2^+_1 ) state at 1.408 MeV Z. Phys. A342 (1992) 17

M. Steinmayer, K.E.G. Löbner, L. Corradi, U. Lenz, P. Pascholati, U. Quade, K. Rudolph, W.K. Schomburg, S.J. Skorka Fission and Gamma-Ray Decay of the^{238}U Shape Isomer Z. Phys. A341 (1992) 145

M. Thoma and E. Nolte Limits on small violation of the Pauli exclusion principle in the primordial nucleosynthesis Phys. Lett. B291 (1992) 484

F. Weber and N.K. Glendenning Applicability of the improved Hartle Method for the Construction of General Relativistic Rotating Neutron Star Models Astrophys. J. 390 (1992) 541

F. Weber and N.K. Glendenning Exact versus Approximate Solution of Einstein’s Equations for Rotating Neutron Stars Phys. Lett. B265 (1991) 1

F. Weber and N.K. Glendenning Impact of the Nuclear Equation of State on Models of Rotating Neutron Stars in: Proceedings of the International Workshop on Unstable Nuclei in Astrophysics, June 7-8, 1991, Tokyo, Japan eds. S. Kubono and T. Kajino, World Scientific 1992

M. Wieland, J. Fernández-Niello, F. Rieß, M. Aïche, A. Chevalier, J. Chevalier, N. Schulz, J.C. Sens, Ch. Briançon, R. Kulessa, E. Ruchowska Side bands in^{219}Ra Phys. Rev. C45 (1992) 1035

M. Wieland, J. Fernández-Niello, B. von Fromberg, F. Rieß, M. Aïche, A. Chevalier, J. Chevalier, N. Schulz, J.C. Sens, Ch. Briançon, E. Ruchowska Search for the I^π=1^− state in^{218}Ra Phys. Rev. C46 (1992) 2628

H.-J. Wollersheim, P. Egelhof, H. Emling, J. Gerl, W. Henning, R. Holzmann, R. Schmidt, R.S. Simon, N. Martin, G. Eckert, Th.W. Elze, K. Stelzer, R. Kulessa, G. Duchene, B. Haas, J.C. Merdinger, J.P. Vivien, J. de Boer, E. Hauber, K. Kaiser, P. von Brentano, R. Reinhardt, R. Wirowski, R. Julin, C. Fahlander, I. Thorslund, H. Kluge Evidence for a Two-Phonon Octupole Vibrational State in^{208}Pb Z. Phys. A341 (1992) 137

– 217 –

K. Zapfe, H. Mairon, J. Stenger, F. Stock, W. Brückner, M. Düren, H.G. Gaul, Z. Moroz, B. Povh, K. Rith, E. Steffens, D. Toporkov, M. Veltri, F. Zetsche, T.A. Shibata, D. Fick, F. Rathmann, G. Graw, P. Schiemenz, W. Haeberli, G. Garvey The Polarized Internal Target for the FILTEX Test Experiment Verhandl. DPG (VI) 27 (1992) 37 E1.3